Author's response to reviews

Title: Understanding and Retention of informed consent process among Parents in a Rural Northern Ghana

Authors:

Abraham R Oduro (aroduro@yahoo.com)
Raymond A Aborigo (raborigo@navrongo.mimcom.net)
Dickson Amugsi (damugsi@navrongo.mimcom.net)
Francis Anto (fanto@navrongo.mimcom.net)
Thomas Anyorigiya (tanyorigiya@navrongo.mimcom.net)
Frank Atuguba (fatuguba@navrongo.mimcom.net)
Abraham Hodgson (ahodgson@navrongo.mimcom.net)
Kwadwo A Koram (kkoram@noguchi.mimcom.net)

Version: 6 Date: 28 May 2008

Author's response to reviews:

We have effected all the formatting requirement requested.

Regards